Author’s response to reviews

Title: Bilio-Entero-Gastrostomy: prospective assessment of a modified biliary reconstruction with facilitated future endoscopic access.

Authors:

Mostafa A Hamad (mostafa_hamad@yahoo.com)
Hosein El-Amin (elamin67@yahoo.com)

Version: 2 Date: 14 October 2011

Author’s response to reviews: see over
Dear Sir,

I am pleased to Re-submit the revised edition of this paper.

The following changes were done according to your request.

1- It was clearly stated in the section of the "methodology" that an informed consent was obtained either from the patient or from his parents.

2- The "trial registration number was added to the abstract section (ISRCTN10135364).

Best regards.

Corresponding Author
Dr. Mostafa A. Hamad
Assistant Professor of Surgery,
Assiut University Hospital,
71516 Assiut, Egypt
e-mail: mostafa_hamad@yahoo.com